Author's response to reviews

Title: Determinants of non attendance to mammography screening in a region with high voluntary private health insurance coverage

Authors:

Magdalena Esteva (mesteva@ibsalut.caib.es)
Joana Ripoll (jripoll@ibsalut.caib.es)
Alfonso Leiva (aleiva@ibsalut.caib.es)
Carmen Sánchez-Contador (csanchezcontador@dgsanita.caib.es)
Francisca Collado (fcollado@dgsanita.caib.es)

Version: 5 Date: 18 September 2008

Author's response to reviews:

Comments to the reviewer Marisa Baré:

1. Reasons for low response to question about attitudes among family and friends has been included in the manuscript in the discussion paragraph on previous experience in breast disease: "The low response rate to that question may have reflected limited knowledge regarding relatives’ and friends’ opinions about the screening program."

2. The word “programme” has been added to the title.

Coments to Dr Mahoney.

The english language has been improved.